Author's response to reviews

Title: Effects of bone marrow cell transplant on thyroid function in an I131-induced low T4 and elevated TSH rat model

Authors:

Gustavo E Guajardo-Salinas (guegsa@itesm.mx)
Juan A Carvajal (A00644163@itesm.mx)
Angel A Gaytan (A00791068@itesm.mx)
Luis Arroyo (A00791784@itesm.mx)
Alberto G Lopez-Reyes (allorey@itesm.mx)
Jose F Islas (chepo3d@itesm.mx)
Beiman G Cano (A00790999@itesm.mx)
Netzahualcoytl Arroyo-Curas (A00790778@itesm.mx)
Alfredo Davalos (A00915358@itesm.mx)
Gloria Madrid (gloria.madrid@gmail.com)
Jorge E Moreno-Cuevas (jemoreno@itesm.mx)

Version: 4 Date: 7 December 2006

Author's response to reviews: see over
Dear Editor:

Thank you very much for reviewing our article, we are happy to hear that the work is of interest for your Journal.

We have reviewed your suggestions and made the required changes:

- First, we have changed figure 4 to only one magnification picture (40x) of each group. We have selected the best and more representative images for group I-IV.
- Also, we have done minor corrections in the written English in order to improve the quality of our manuscript.

The changes on the manuscript have been minimal. There were no major corrections requested. Only minor corrections were suggested and they were corrected as indicated.

We will be happy to address any other concerns and we will be waiting for your final decision.

Again, thank you for your attentions.

Sincerely,

Gustavo E. Guajardo MD et al.